Author's response to reviews

Title: Predicting the onset and persistence of episodes of depression in primary health care: the PREDICT-Spain study. Methodology.

Authors:

Juan A Bellon (JABELLON@terra.es)
Berta Moreno-Kustner (bertamk@uma.es)
Francisco Torres-Gonzalez (ftorres@ugr.es)
Carmen Monton-Franco (carmenmonton@able.es)
Maria J GildeGomez-Barragan (jggomez@riojasalud.es)
Marta Sanchez-Celaya (mscelaya@gmail.com)
Miguel A Diaz-Barreiros (mdiazba@comlp.es)
Catalina Vicens (domcatrin@telefonica.net)
Juan D Luna (jdluna@ugr.es)
Jorge A Cervilla (jacb@ugr.es)
Blanca Gutierrez (blancag@ugr.es)
Maria T Martinez-Canavate (tmcanavate@samfyc.es)
Barbara Olivan-Blazquez (boli@aragob.es)
Ana Vazquez-Medrano (avazquez@riojasalud.es)
Maria S Sanchez-Artiaga (malus209@hotmail.com)
Sebastia March (sma@bsalut.caib.es)
Emma Motrico (emm@uma.es)
Victor M Ruiz-Garcia (vmruiz@uma.es)
Paulette R Brangier-Waingber (paulette.brangier@gmail.es)
Irwin Nazareth (IN@gprf.mrc.ac.uk)
Michael King (m.king@medsch.ucl.ac.uk)
María M Muñoz-García (marmg@ugr.es)

Version: 2 Date: 17 March 2008

Author's response to reviews:

Facultad de Medicina
Departamento de Medicina Preventiva y Salud Pública
Universidad de Málaga
Campus de Teatinos s/n
29071. Málaga (SPAIN)

March 25th 2008
BMC Public Health
Dear Annabel Phillips,

We have done a new version of our paper MS:1583737518180283, ¿Predicting
the onset and persistence of episodes of depression in primary health care. The PREDICT-Spain study: methodology. We have removed the baseline data and the data presented for the 6 and 12 month follow up period and, considering your previous permission, we have included the reliability data for the questionnaire. If you accept our paper to be published, we would like to pay before 30 March because we have to close our grant for that date.

We look forward to hearing from you.

Yours sincerely:

Juan Ángel Bellón